**Author's response to reviews**

**Title:** The short international physical activity questionnaire: cross-cultural adaptation, validation and reliability of the Hausa language version in Nigeria

**Authors:**

Adewale L Oyeyemi (alaoyeyemi@yahoo.com)
Adetoyeje Y Oyeyemi (adeoyeyemi@aol.com)
Babatunde O Adegoke (badegoke@comui.edu.ng)
Fatima O Oyetoke (foyetoke@yahoo.com)
Habeeb N Aliyu (habibaliyuwd@yahoo.com)
Salamatu U Aliyu (ptsalualiyu@yahoo.com)
Adamu A Rufai (adamuarufai@gmail.com)

**Version:** 7  **Date:** 21 November 2011

**Author's response to reviews:**

For consistency, we have changed male to men and female to women under gender in table 2. We have also effected this change on page 17, first paragraph.